Author's response to reviews

Title: Return to fertility after extended chemical castration with a GnRH antagonist

Authors:

Janusz W Kostanski (janusz.kostanski@bms.com)  
Ge Jiang (gjian0@uky.edu)  
Bhas A Dani (bdani@inhale.com)  
Santos B Murty (smurt1@uky.edu)  
Wei Qiu (wqiu2@uky.edu)  
Bruce Schrier (bschrier@oakwoodlabs.com)  
B.C. Thanoo (thanoo@oakwoodlabs.com)  
Patrick P DeLuca (ppdelu1@uky.edu)

Version: 5 Date: 29 Oct 2001

The following revisions were made on October 29, 2001:
1. Unnecessary capitals were removed from the title.
2. Abbreviated title and keywords have been removed from the cover page.
3. Figures 2 and 3 were reformatted.